Author’s response to reviews

Title: Development and initial validation of an instrument to measure novice nurses’ perceived ability to provide care in acute situations – PCAS

Authors:

Anders Sterner (anders.sterner@hb.se)

Emma Säfström (emma.safstrom@liu.se)

Lina Palmér (lina.palmer@hb.se)

Nerrolyn Ramstrand (nerrolyn.ramstrand@ju.se)

Magnus Andersson Hagiwara (magnus.hagiwara@hb.se)

Version: 3 Date: 07 Feb 2020

Author’s response to reviews:

Thank you for considering our research for publication.

1. Please clearly state whether the need for ethics approval was waived by national/regional guideline, with the references you had provided.

Inserted the statement: Ethical approval for this study was waived by the national Ethical Review Act 2003:460 in the Ethics approval and consent to participate section

I hope that the edit have been made to your satisfaction.

Sincerely Yours

Anders Sterner